		EO-1 Weekly Report
	Earth Observing – 1

	March 02, 2016 – March 08, 2016

	DOY
	062
	–
	068

	Mission Day
	5630
	–
	5636


EO-1 SPACECRAFT

Instruments
· Scheduled 117 science Data Collection Events (DCEs)
· Loss of [1] DCE, see the operational discrepancies section below for more details. 
· Performed Instrument Calibration 
· Conducted Ali Internal Cal 1 on DOY 067/01:29z

EO-1 MISSION OPERATIONS CENTER
Real-Time
· Maintaining nominal spacecraft state of health
· Tested USN upgraded IONet successfully 
· Prepared for the eclipse on DOY 069
· AS1/AS3 engineering supports are in the schedule for this past week and future weeks
· AS1 
· S-band supports have been successful  
· X-band supports have been successful 
· AS3 
· S-band supports have been successful 
· X-band supports have been successful 
· Working on upgrading the FD windows XP machines to windows 7
· See FD section of this report for more details 
· Working on combined Hyperion/ALI slow scan lunar calibration and multi-orbit lunar calibration
· Working to set up automated GPS telemetry capture to work with FD
· Manual set-up is complete, will create FD scenario to use GPS data
· Testing Overdraw after update in downtime of FD systems

Mission Planning
· Blocked off ASE activities for 2-orbits on DOY 069 during the eclipse to conserve power on the spacecraft 
· Verifying and selecting AS1/AS3 S- and S/X- band  supports
· Adding in AS 1/3 ground stations to ASPEN
· Working with the USAK04 personnel to bring test passes with that ground station

Flight Dynamics 
· Verified orbits to have ASE blocked out for DOY 069 spacecraft lunar eclipse
· One orbit blocked with 94% occultation during mid spacecraft day
· One orbit fine with 9 min occultation at start of day at 22% and going down from there
· AS1 and AS3 are now input into all FD products with masking file and station ID
· Testing AS1 and AS3 tracking data
· AS1 tracking data is having issues, the station is replacing a part in a few weeks that should fix this
· [bookmark: _GoBack]AS3 tracking data is looking good so far and is in line with our normal tracking data
· Upgrading Waldorf2 to Windows 7
· Waldorf2 is now in parallel testing with Statler2 for validation
· SA fixed a problem that has prevented SSH from working on Windows 7 since the upgrade
· Comparing routine FD products (so far all products match)
· Testing Lunar Calibration products
· Testing DeltaV generation products
· Routine products created on Waldorf2 were delivered from Statler2 without issue
· Will attempt delivery of FD products from Waldorf2 next
· Add GPS OD scenario to determine the orbit solely using the GPS telemetry data
· Designing a multi-orbit lunar calibration sequence since the nominal is barely fitting into spacecraft night
· Working to create a new slow scan combined Hyperion/ALI and fit it onto the warp for future off-angle collects
· Contacted AGI to update the STK PODS tables file. If this file wasn’t updated, the EO-1 MOC wouldn’t have been able to continue performing Orbit Determination. 
· AGI will not update the file any longer
· Received scripts from them that they use to create the tables and are in process to get them working correctly.
· Verifying rough draft of delta-v planning procedures

Central File Hub
· Phase one tests are ready for outside entity delivery.
· USAK03, AS1, AS3 all deliver to file hubs and its working fine
· SGS and WGS tracking data are waiting for a break in WALLOPS activities to begin testing.
· Phase 2 FD scripts have no changes from Phase 1 scripts
· All non-FD scripts are implemented Phase 2 
· Beginning Phase 3 work on deliveries for the central file hub
· Central file hub delivery structure set
· Passwordless ssh set up from FD to eo1out and eo1in to and from eo1out
· Script to put FD products onto file hubs is complete and ready for use
· Script to sort FD products into destination directories is complete and ready for use
· Base delivery script completed
· Designing delivery code architecture
· Reviewing current deliveries for products, emails, and destinations
· Delivery will be completed via FD computers as is the current standard

System Administration
· Participated in ESMO System Security Plan Control Implementation meeting
·  Worked to recover a file for FOT
· Tried to upgrade firmware on Colonel to fix a medium vulnerability.
· Requested renewal of three MSGRS requests.
· Worked on correcting deficiencies in the Waldorf2 backup
· Completed MOC password change
· Participated in weekly security tag up
· Attended presentation on 4 certificate PIV
· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, regional logs server, and a full backup of the MOC CNE computer.

GROUND AND SPACE NETWORK
Station Downtimes 	
· None to report at this time

[bookmark: _MON_1387176949][bookmark: _MON_1388382035][bookmark: _MON_1388382145][bookmark: _MON_1388384576][bookmark: _MON_1388385083][bookmark: _MON_1388993005][bookmark: _MON_1389598124][bookmark: _MON_1390809671][bookmark: _MON_1390809790][bookmark: _MON_1391499119][bookmark: _MON_1391501949][bookmark: _MON_1392188921][bookmark: _MON_1392615265][bookmark: _MON_1393314853][bookmark: _MON_1393834790][bookmark: _MON_1394437595][bookmark: _MON_1394437598][bookmark: _MON_1394437644][bookmark: _MON_1395039355][bookmark: _MON_1395831044][bookmark: _MON_1396332066][bookmark: _MON_1396332146][bookmark: _MON_1396332643][bookmark: _MON_1396332666][bookmark: _MON_1396332684][bookmark: _MON_1396332702][bookmark: _MON_1396862017][bookmark: _MON_1397634356][bookmark: _MON_1397634698]Operational Discrepancies:
· Downlink: 2016:064:10:00:30 2016:064:10:11:49 SG1
· Problem:  CDS ID# 263510 GN: SG-1 - EO-1: Antenna lost signal at high angle and oscillated for 30 seconds.
· Result:   Loss of one [1] scene




UPCOMING EVENTS
· Testing and validation of FD systems on Windows 7, ASIST systems on RedHat upgrade
· Testing with AS1 and AS3
· Continuation of Phase 1/2 FD, waiting on WALLOPS support 
· Moving on to Phase 3 for central file hub, deliveries
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Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

2-Mar-16 62 7 5 0 0 12

3-Mar-16 63 6 7 0 0 13

4-Mar-16 64 8 2 0 0 10

5-Mar-16 65 8 4 0 0 12

6-Mar-16 66 6 3 0 0 9

7-Mar-16 67 4 6 0 0 10

8-Mar-16 68 5 6 0 0 11

44 33 0 0 77 Weekly Totals
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		2-Mar-16		62		7		5		0		0		12

		3-Mar-16		63		6		7		0		0		13

		4-Mar-16		64		8		2		0		0		10

		5-Mar-16		65		8		4		0		0		12

		6-Mar-16		66		6		3		0		0		9

		7-Mar-16		67		4		6		0		0		10

		8-Mar-16		68		5		6		0		0		11

		Weekly Totals				44		33		0		0		77
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